Author's response to reviews

Title: Urgent endovascular ligature of ruptured splenic artery pseudo-aneurysm (PSA) as a complication of acute pancreatitis in a patient with impaired renal function. - Case Presentation-

Authors:

Anna Maria Ierardi (amierardi@yahoo.it)
Mario Petrillo (mario.petrillo@gmail.com)
Chiara Floridi (chiara.floridi@gmail.com)
Federico Fontana (fede.fontana@libero.it)
Ejona Duka (ejonaduka@hotmail.com)
Gianpaolo Carrafiello (gcarraf@gmail.com)

Version: 2
Date: 12 August 2014

Author's response to reviews:

Change requested were made as follows:

2. Was reformatted the title page as page 1.

3. Was restructured the Abstract section on page 2.

4. Was changed the ‘Case Report’ section header to ‘Case Presentation’.

5. Was included the Ethnicity of the patient in the Abstract and Case presentation sections.

6. Was reformatted the Consent statement.

7. Were not included footnotes are not allowed, but endnotes are permitted.

8. Was included include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.

9. Was included an Authors’ Contributions section at the end of the manuscript, before the reference list.

10. Was not included any an acknowledgement section because was not available any kind of funding.